

Citation: Khanolkar AR, Chaturvedi N, Kuan V, Davis D, Hughes A, Richards M, et al. (2021) Socioeconomic inequalities in prevalence and development of multimorbidity across adulthood: A longitudinal analysis of the MRC 1946 National Survey of Health and Development in the UK. PLoS Med 18(9): e1003775. https://doi.org/10.1371/ journal.pmed.1003775

Academic Editor: Sanjay Basu, Harvard Medical School, UNITED STATES

Received: November 2, 2020

Accepted: August 19, 2021

Published: September 14, 2021

Copyright: © 2021 Khanolkar et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly as sharing of the 1946 MRC National Survey of Health & Development (NSHD) data is dependent on the project being approved by the NSHD Data Sharing Committee, a data sharing agreement being in place between UCL and the academic institution that employs the researcher, and LHA resources being available to meet the requests for data sharing. Additionally, all proposals to use the NSHD data must support and RESEARCH ARTICLE

Socioeconomic inequalities in prevalence and development of multimorbidity across adulthood: A longitudinal analysis of the MRC 1946 National Survey of Health and Development in the UK

Amal R. Khanolkar^{1,2}, Nishi Chaturvedi¹, Valerie Kuan³, Daniel Davis¹, Alun Hughes¹, Marcus Richards¹, David <mark>Bann⁴, Praveetha</mark> Patalay^{1,4}*

1 MRC Unit for Lifelong Health and Ageing at UCL, London, United Kingdom, 2 Department of Global Public Health, Karolinska Institutet, Stockholm, Sweden, 3 Institute of Health Informatics, UCL, London, United Kingdom, 4 Centre for Longitudinal Studies, UCL, London, United Kingdom

* p.patalay@ucl.ac.uk

Abstract

Background

We aimed to estimate multimorbidity trajectories and quantify socioeconomic inequalities based on childhood and adulthood socioeconomic position (SEP) in the risks and rates of multimorbidity accumulation across adulthood.

Methods and findings

Participants from the UK 1946 National Survey of Health and Development (NSHD) birth cohort study who attended the age 36 years assessment in 1982 and any one of the follow-up assessments at ages 43, 53, 63, and 69 years (N = 3,723,51% males). Information on 18 health conditions was based on a combination of self-report, biomarkers, health records, and prescribed medications. We estimated multimorbidity trajectories and delineated socio-economic inequalities (based on childhood and adulthood social class and highest education) in multimorbidity at each age and in longitudinal trajectories.

Multimorbidity increased with age (0.7 conditions at 36 years to 3.7 at 69 years). Multimorbidity accumulation was nonlinear, accelerating with age at the rate of 0.08 conditions/ year (95% CI 0.07 to 0.09, p < 0.001) at 36 to 43 years to 0.19 conditions/year (95% CI 0.18 to 0.20, p < 0.001) at 63 to 69 years. At all ages, the most socioeconomically disadvantaged had 1.2 to 1.4 times greater number of conditions on average compared to the most advantaged. The most disadvantaged by each socioeconomic indicator experienced an additional 0.39 conditions (childhood social class), 0.83 (adult social class), and 1.08 conditions (adult education) at age 69 years, independent of all other socioeconomic indicators. Adverse adulthood SEP was associated with more rapid accumulation of multimorbidity, resulting in 0.49 excess conditions in partly/unskilled compared to professional/intermediate individuals between 63 and 69 years. Disadvantaged childhood social class, independently of adhere to the core principles of data sharing with the Medical Research Council (MRC). For more information on the NSHD, access to data & collaborations, please see http://www.nshd.mrc.ac. uk/data/data-sharing/ or contact mrclha. enquiries@ucl.ac.uk Further information can also be accessed at: https://skylark.ucl.ac.uk.

Funding: This study was funded by the following grants: This work was supported by the UK Medical Research Council which provides core funding for the MRC Unit For Lifelong Health and Ageing at UCL and the MRC National Survey of Health and Development (MC_UU_00019/1, MC_UU_00019/3). Amal Khanolkar was supported by grant from the Wellcome Trust (ISSF3/H17RCO/ NG1) and MRC (MC_UU_00019/3). DB is supported by the Economic and Social Research Council (grant number ES/M001660/1) and The Academy of Medical Sciences/Wellcome Trust ("Springboard Health of the Public in 2040" award: HOP001/1025). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the following authors of this manuscript have competing interests: NC is remunerated for her membership of a data safety and monitoring committee of a trial sponsored by AstraZeneca. The remaining authors have no competing interests to declare. All other authors have declared that no competing interests exist.

Abbreviations: MAR, missing at random; NSHD, National Survey of Health and Development; SEP, socioeconomic position.

adulthood SEP, was associated with accelerated multimorbidity trajectories from age 53 years onwards.

Study limitations include that the NSHD cohort is composed of individuals of white European heritage only, and findings may not be generalizable to the non-white British population of the same generation and did not account for other important dimensions of SEP such as income and wealth.

Conclusions

In this study, we found that socioeconomically disadvantaged individuals have earlier onset and more rapid accumulation of multimorbidity resulting in widening inequalities into old age, with independent contributions from both childhood and adulthood SEP.

Author summary

Why was this study done?

- Multimorbidity—2 or more chronic conditions in an individual—has several consequences including reduced quality of life, reduced life expectancy, and complex healthcare needs.
- Socioeconomic disadvantage is associated with both earlier onset (at younger ages) and greater burden of multimorbidity. However, less is known about how multimorbidity develops across adulthood and into older ages in the same individuals and differences in its development by childhood and adulthood socioeconomic position (SEP).

What did the researchers do and find?

- Longitudinal multimorbidity trajectories were estimated between ages 36 and 69 years, and whether these trajectories differed by childhood (father's social class) and adulthood SEP (participant's own social class and educational level).
- We found that throughout follow-up, the most socioeconomically disadvantaged had 1.2 to 1.4 times greater number of health conditions on average compared to the most advantaged.
- On average, multimorbidity increased across adulthood and into older ages in both socioeconomically disadvantaged and advantaged groups. However, the rate at which multimorbidity increased across adulthood differed by SEP, with those from the most disadvantaged backgrounds having worse trajectories. In other words, adverse SEP was associated with more rapid accumulation of multimorbidity.
- Childhood and adulthood SEP were independently associated with multimorbidity across adulthood. This indicates that childhood socioeconomic disadvantage is still evident in later life over and above adulthood socioeconomic circumstances.

What do these findings mean?

- The independent and long-lasting shadow of childhood SEP on the experience and the rate of accumulation of multimorbidities in later life highlight the need for interventions early in life to minimise inequalities in later life health.
- This calls for population-based interventions in early life and through the lifecourse to
 reduce the impact of childhood and adulthood inequalities, along with better access and
 delivery of healthcare for the more vulnerable to help reduce the burden of
 multimorbidity.

Introduction

The management of long-term chronic health conditions poses a serious challenge for public healthcare systems worldwide [1,2]. Conditions associated with unhealthy lifestyles, such as obesity, hypertension, diabetes, and coronary heart disease are increasingly common, while greater life expectancies mean individuals live longer with chronic disease. These conditions rarely occur in isolation, hence multimorbidity—2 or more chronic conditions in an individual—is a rising consequence [3]. Multimorbidity is associated with poorer quality of life, decreased work productivity, increased use of ambulatory and inpatient healthcare, polypharmacy, increased functional limitations, and mortality [4–9]. Socioeconomic inequalities in health are well established, and socioeconomic disadvantage is associated with both a greater burden of multimorbidity and earlier onset [2,10–22].

Most studies on the epidemiology of multimorbidity, and specifically those that have investigated socioeconomic inequalities, have largely been cross-sectional in design [2,4,12,13,15,17,23]. The few longitudinal studies vary substantially in design (cohorts following same individuals and serial cross-sectional surveys) and follow-up time (from 5 to 25 years) [14,16,24–26], and to our knowledge, no study has examined the longitudinal development of multimorbidity, including rate of accumulation, in the same individuals and both sexes across adulthood. Additionally, the longitudinal effects of early and later life socioeconomic position (SEP) on prevalence and development of multimorbidity is not known.

Independent effects of socioeconomic disadvantage in childhood on multimorbidity development across the lifecourse are not known. This is an important gap in knowledge, as socioeconomic disadvantage in childhood has lasting impacts on health, independently of adult socioeconomic circumstances [27,28]. Excluding childhood socioeconomic circumstances might underestimate the extent of the impact of socioeconomic disadvantage across the lifecourse.

Further, few studies to date have investigated multimorbidity development in the same individuals across the lifecourse, and no studies investigated its association with lifetime SEP, as this requires individual-level population-based longitudinal data over several decades. Given ageing-related processes and health deterioration, it is expected that the rate of multimorbidity accumulation will accelerate over the lifecourse. Any socioeconomic differentials in the rate of disorder accumulation are important to establish as they help uncover whether the inequalities observed at older ages are driven by earlier onset, faster accumulation, or both.

This study examines multimorbidity development across adulthood and into older ages and investigates in detail socioeconomic inequalities in multimorbidity trajectories in a prospective nationally representative birth cohort study. We estimated multimorbidity development in the same individuals over time and, importantly, estimate the independent effects of childhood and adulthood socioeconomic circumstances on multimorbidity prevalence and rate of accumulation.

Methods

Study participants and setting

This study analysed data from participants in the MRC 1946 National Survey of Health and Development (NSHD) [29]. The NSHD includes a socially stratified and nationally representative sample of 5,362 participants born in 1 week in March 1946 in England, Scotland, and Wales and recruited at birth. Extensive social, anthropometric, medical, lifestyle, and biological data were collected either during home visits or at clinical research centres in later years by professional interviewers and research nurses. More information on the study is available from the study's website: https://www.nshd.mrc.ac.uk/. Each sweep of data collection has the required ethics approvals, and study members provide their consent for each assessment sweep.

The analysis sample for this prospective longitudinal study included participants who attended the age 36 years assessment in 1982 and any one of the follow-up assessments at ages 43 in 1989, 53 in 1999, 63 in 2009, and 69 years in 2015 (total N = 3,723, including 1,843 women [49.5%]). Participants lost to follow-up (due to death or emigration) between ages 36 and 63 were excluded from the study sample.

Attrition: Of the original 5,362 cohort participants, 3,723 participants were included in this analysis. By age 63 years, 718 (13.4%) had died, 594 (11.1%) had previously withdrawn from the study, 567 (10.6%) lived abroad, and 320 (5.9%) had been untraceable for more than 10 years.

Conditions

Data for 18 selected common chronic conditions on which consistent and reliable information was collected across the adulthood sweeps are the focus of this study [30]. These included diabetes, dyslipidaemia, hypertension, obesity, coronary heart disease, stroke, cancer, anaemia, respiratory disorders, kidney disorders, gastrointestinal disorders, skin disorders, osteoarthritis, rheumatoid arthritis, Parkinson disease, epilepsy, depression, and psychosis. Including conditions with consistently collected data across the 5 sweeps would aid in estimating total multimorbidity score at each time point. Data sources, sweeps at which information is available and definitions for each of the 18 conditions, are listed in Table A in S1 File.

Chronicity

We used the Delphi technique [31]—a structured communication tool—to give an informed consensus on which conditions would be considered chronic (i.e., once a participant was diagnosed with a condition then he/she would be considered to have the condition throughout follow-up). The Delphi team consisted of 10 clinicians (of ranging specialities including general practice, cardiovascular, geriatrics, and respiratory medicine) who received the same form that listed the 18 conditions, how they were measured, and a column for them to indicate which ones they would consider chronic (i.e., to be carried forward; Table B in S1 File). Each clinician was requested to indicate their decision along with a brief explanation supporting their decision. All responses were collated and examined in an Excel sheet. We chose the majority decision for each condition on whether it should be considered chronic or not.

Multimorbidity

All 18 conditions were coded as binary variables indicating whether a participant had a condition or not at each sweep. The multimorbidity score in this paper is defined as the unweighted sum of the number of conditions (out of 18 conditions) accumulated by an individual [2,14].

For those conditions considered chronic (like diabetes or hypertension), a participant was coded to have the condition throughout follow-up once diagnosed (Table A in S1 File). For all participants, we calculated a multimorbidity score at each age (i.e., across the 5 age sweeps), which indicated the number of conditions at each time point, and this multimorbidity score was the main outcome.

Socioeconomic position (SEP)

Childhood SEP was based on the father's occupational social class, reported at age 11. We used 2 measures of SEP in adulthood: educational level attained at age 33 and occupational social class ascertained at age 43. Occupational social class was coded into 4 categories: (1) Professional and intermediate (reference group in analysis); (2) Nonmanual; (3) Manual; and (4) Partly skilled/unskilled. Educational levels were ascertained by the Burnham classification. The original 9 categories were collapsed into 3 groups for analysis: (1) None or vocational courses; (2) School leaving certificate (GCE "O" and "A" levels); and (3) Degree level and higher (university education, reference group).

Statistical analysis

We used frequencies, percentages (prevalence of each condition at all age sweeps for the full study sample), cross-tabulations (mean differences in multimorbidity score across the 5 age sweeps by socioeconomic indicators), and graphical display for descriptive analysis. To visualise prevalence and associations between the conditions that constituted the multimorbidity score at each age, networks were constructed using the R package "igraph" (conditions represented by nodes with sizes proportional to prevalence). These networks effectively illustrate the changing makeup of the total multimorbidity score at each age. Partial correlations between each of the 18 conditions are represented as "edges" in networks (networks only included those edges with partial correlations with absolute values >0.05 and *p*-values <0.05). The width of an edge was proportional to the strength of the partial correlation between 2 conditions (with positive partial correlations >0.05 represented by grey edges and negative correlations <-0.05 shown in red).

Socioeconomic patterning in multimorbidity at each of the 5 age sweeps was analysed using multivariable linear regression. We first ran models to test individual associations between each of the 3 indicators and the multimorbidity score. The fourth model was mutually adjusted for all 3 socioeconomic indicators in order to assess each of their associations independent of other indicators. All models were adjusted for sex.

Longitudinal change in multimorbidity trajectories were analysed using linear spline mixed-effects modelling. This accounts for repeated measurements (or clustering) of multi-morbidity scores within individuals over time and individual-level deviations from an overall average trajectory, with a nonlinear specification of the average trajectory using linear splines [32]. Given the data structure—cohort members are of the same age at each time point—knots were assigned at ages 43, 53, 63, and 69 years giving 4 splines corresponding to ages 36 to 43, 43 to 53, 53 to 63, and 63 to 69 years. The model with a random intercept plus random slopes for the individual spline coefficients had a superior fit and was used in all subsequent model-ling. This breaks up the average change in multimorbidity score into separate slopes for each of the 4 spline variables representing an age band (each slope coefficient indicating rate of

change in multimorbidity score per year in each period). We used linear spline mixed-effects modelling to first estimate the population-average multimorbidity trajectory for the full sample and by sex. The model for multimorbidity trajectories by sex included interaction terms between sex and the spline variables for ages 36 to 43, 43 to 53, and 53 to 63 years. We then ran 4 models, each one having a different socioeconomic exposure of interest plus sex: Model 1: childhood social class; Model 2: adulthood social class; Model 3: adulthood educational level; and Model 4: mutual adjustment for all 3 socioeconomic indicators.

To examine whether the slope of multimorbidity trajectories varied by social class categories over time within each of the 4 spline variables (i.e., whether inequalities in the rate of multimorbidity accumulation changed change over time), we tested for interactions between childhood or adulthood social class and the individual spline variables. Only those interactions terms that were significant (p < 0.05) were subsequently included in the model. The model for adulthood social class included interactions with all 4 spline variables (as all 4 interactions were significant, p < 0.05), whereas the one for childhood social class included interactions with the ages 53 to 63 years and 63 to 69-years' variables only. These models were mutually adjusted for other socioeconomic indicators. Interactions were significant, and multimorbidity trajectories were plotted at the group level (childhood and adulthood social class separately) for visualisation based on model 4 above. Interactions between spline variables and education were not significant and are not shown.

Missing data across the 5 age sweeps were addressed using multiple imputation assuming missing at random (MAR) using 25 imputed datasets. The MAR mechanism implies that systematic differences between the missing values and the observed values can be explained by observed data [33], which is a plausible assumption in the British birth cohorts given the rich data available from birth [34]. Additionally, the main purpose was to impute missing data in health conditions across the 5 sweeps. The imputed data are more likely to be closer to the true values (i.e., predicted with greater precision), as many of the health conditions are correlated longitudinally further strengthening the imputation model. Information on the 3 socioeconomic indicators was available for more >94% of study participants. Frequencies and distributions of nonimputed and imputed variables were largely similar for most conditions (Table C in S1 File). All analyses including the descriptive analysis were run on the imputed sample (N = 3,723). All analyses were conducted using Stata version 15 (StataCorp LP, College Station, Texas). A completed STROBE checklist is provided as a separate supplementary file (S1 STROBE Checklist). Analyses were prespecified, but no written analysis plan is available.

Results

Table 1 shows the prevalence of the 18 conditions across the 5 ages. We found that the prevalence of most conditions increased with age (for example, obesity and hypertension increased from 7% and 6% at age 36 to 34% and 52% at age 69 years, respectively). The proportion with no conditions decreased progressively from 52% at age 36 to 7% at age 69 years (Fig 1). Fig 2 displays results from network analysis with node size representing relative prevalence of a condition and edges representing associations between conditions at each of the 5 ages. These networks illustrate both the relative increase in prevalence of disorders over time and the greater interconnectivity between conditions with age. For example, prevalence of hypertension increases with age, and its association with dyslipidaemia gets stronger over time. In contrast, although more individuals have cancer over time, cancer is not independently associated with experiencing any of the other conditions at most ages (indicated by no edges between cancer and other conditions). Similarly, skin conditions are independent of most conditions over time, apart from respiratory conditions.

Condition	Age 36	Age 43	Age 53	Age 63	Age 69	
Obesity	6.9 (6-7.7)	12.8 (11.6–13.9)	25.0 (23.2–26.6)	32.2 (30-34.4)	34.4 (32–36.7)	
Hypertension	6.3 (5.4–7.1)	13.3 (12.1–14.5)	32.4 (30.7–34.1)	44.8 (42.9–46.6)	52.0 (50–53.9)	
Dyslipidaemia	N.A.	N.A.	9.1 (7.9–10.2)	32.2 (30–34.3)	47.8 (45.4–50)	
Diabetes	N.A.	1.5 (1-1.9)	3.2 (2.5–3.8)	10.7 (9.4–11.9)	16.3 (14.8–17.7)	
CHD	N.A.	3.8 (3.1–4.5)	11.6 (10.2–12.9)	20.2 (18.5–22)	22.8 (20.9–24.5)	
Stroke	N.A.	N.A.	2.5 (1.4–3.6)	6.4 (4.7–8.2)	10.3 (8.6–11.9)	
Osteoarthritis	N.A.	N.A.	N.A.	9.4 (7.8–10.9)	27.5 (25.1–29.8)	
Rheumatoid arthritis	N.A.	N.A.	1.3 (0.5–2.2)	4.2 (2.5–5.9)	9.7 (7.3–12)	
Anaemia	N.A.	N.A.	9.0 (7.8–10.2)	6.8 (5.2–8.5)	12.5 (10.4–14.6)	
Skin disorders	10.8 (9.7–11.8)	18.6 (17.3–20)	19.1 (17.6–20.5)	N.A.	16 (14.1–18)	
Respiratory disorders	16.1 (14.8–17.4)	28.4 (26.8-30)	17.5 (15.9–19)	20.4 (18.7–22.1)	25.6 (23.7–27.5)	
Gastrointestinal disorders	2.3 (1.6–2.8)	8.9 (7.8–9.8)	8.0 (6.7–9.1)	21.0 (18.6–22.7)	28.3 (25.1–21.5)	
Kidney disorders	4.9 (4.2–5.6)	13.8 (12.6–14.9)	13.3 (11.8–14.8)	N.A.	5.0 (2.8–7.2)	
Parkinson disease	N.A.	N.A.	N.A.	2.6 (1.2-4)	5.5 (3.8–7.1)	
Cancer	0.6 (0.3–0.8)	1.7 (1.3-2.1)	5.6 (4.8-6.3)	14.2 (13.1–15.3)	19.5 (18.3–20.9)	
Epilepsy	1.5 (0.9–1.9)	2.1 (1.4–2.7)	4.2 (2.9–5.4)	3.0 (2.1-4)	8.3 (6.7–9.9)	
Depression	18.4 (16.9–19.7)	16.4 (15.2–17.7)	23.0 (21.3–24.7)	2 3.4 (21.7–25.1)	24.5 (22.7–26.9)	
Psychotic disorders	0.9 (0.5–1.4)	2.0 (1.4–2.7)	3.8 (2.6–5.1)	6.0 (4.4–7.6)	8.0 (6.2–9.8)	
Number of conditions (mean)	0.7 (0.6–0.7)	1.2 (1.1–1.3)	1.9 (1.8–2.0)	2.6 (2.5–2.7)	3.7 (3.6–3.9)	

Table 1. Prevalence (%) of conditions and multimorbidity at each of the 5 age sweeps in 3,723 participants from the 1946 MRC NSHD.

N.A., not applicable—indicates that data for that condition were not collected at the particular age. CHD, coronary heart disease; NSHD, National Survey of Health and Development.

https://doi.org/10.1371/journal.pmed.1003775.t001

We found that, at each age sweep, socioeconomic disadvantage was associated with higher multimorbidity scores (Fig 3, Table 2, and Table F in S1 File). The mean difference in number of conditions between the most and least disadvantaged was observed with childhood and adulthood social class, and with educational level at all ages, and got progressively larger with age. At all ages, the most socioeconomically disadvantaged had a greater number of conditions on average compared to the most advantaged with the largest difference observed at age 69 years with 1.4 more conditions (for instance, 0.60 versus 0.85 at age 36, 1.67 versus 2.28 at age 53, and 3.24 versus 4.59 at age 69 years for professional/intermediate versus partly skilled/ unskilled adulthood social class, respectively; Table F in S1 File). Compared with the most





https://doi.org/10.1371/journal.pmed.1003775.g001





Figure Note: Larger node size (circles) indicate greater relative prevalence of a condition. Thicker edges (lines connecting two conditions or nodes) indicate larger partial correlation coefficients.

Fig 2. Network analysis illustrating relative prevalence (node size) of conditions and adjusted associations (edges) at each age in 3,723 participants from the 1946 MRC NSHD. CHD, coronary heart disease; NSHD, National Survey of Health and Development.

https://doi.org/10.1371/journal.pmed.1003775.g002



A. Childhood social class





C. Adulthood educational level



Fig 3. Mean number of morbidities at each age by socioeconomic indicators in 3,723 participants from the 1946 MRC NSHD. GCE, General Certificate of Education; NSHD, National Survey of Health and Development.

https://doi.org/10.1371/journal.pmed.1003775.g003

	Change in multimorbidity score ^a									
Age 36		Age 43		Age 53	Age 53		Age 63			
Covariates	β	95% CI	β	95% CI	β	95% CI	β	95% CI	β	95% CI
Childhood social class										
Professional/intermediate	Ref		Ref		Ref		Ref		Ref	
Skilled nonmanual	0.03	-0.07,0.12	0.08	-0.05,0.21	-0.01	-0.18,0.17	0.11	-0.13,0.35	0.18	-0.13,0.50
Skilled manual	0.05	-0.04,0.13	0.05	-0.07,0.16	0.09	-0.07,0.24	0.32	0.12,0.52	0.39	0.10,0.68
Partly skilled/unskilled	0.03	-0.06,0.12	0.07	-0.06,0.20	0.04	-0.14,0.22	0.25	0.00,0.50	0.39	0.06,0.72
Adulthood social class										
Professional/intermediate	Ref		Ref		Ref		Ref		Ref	
Skilled nonmanual	0.04	-0.04,0.13	0.03	-0.09,0.15	0.12	-0.04,0.27	0.05	-0.15,0.25	0.11	-0.17,0.39
Skilled manual	0.05	-0.04,0.15	0.03	-0.10,0.16	0.13	-0.05, <mark>0.3</mark> 1	0.24	0.02,0.46	0.33	0.03,0.63
Partly skilled/unskilled	0.18	0.08,0.28	0.17	0.03,0.32	0.43	0.24,0.62	0.54	0.2 <mark>8,0</mark> .80	0.83	0.45,1.20
Educational attainment										
University degree	Ref				Ref		Ref		Ref	
GCE/school leaving certificate	-0.01	-0.11,0.09	-0.05	-0.18,0.09	0.02	-0.18,0.22	0.14	-0.10,0.38	0.35	0.03,0.67
None	0.06	-0.07,0.18	0.13	-0.04,0.30	0.28	0.04,0.53	0.62	0.32,0.91	1.08	0.67,1.49
Sex										
Men	Ref		Ref		Ref		Ref		Ref	
Women	0.09	0.02,0.15	0.15	0.06,0 <mark>.24</mark>	0.06	-0.07,0.18	-0.11	-0.27,0.06	-0.24	-0.46,-0.03

Table 2. Results from linear regression modelling assessing associations between SEP and multimorbidity (unweighted sum of the number of health conditions) in 3,723 participants from the 1946 MRC NSHD.

^aAll models adjusted for sex; coefficients in bold indicate 95% confidence intervals that do not include zero. GCE, General Certificate of Education; NSHD, National Survey of Health and Development; SEP, socioeconomic position.

https://doi.org/10.1371/journal.pmed.1003775.t002

advantaged group (professional/intermediate), partly skilled/unskilled individuals had progressively higher multimorbidity scores across adulthood (Table 2; age 36 years: 0.18 [0.08 to 0.28], age 69 years: 0.83 [045 to 1.20]; and Fig A in S1 File). Socioeconomic disadvantage in childhood and in adulthood were independently associated with higher levels of multimorbidity at ages 63 and 69 years (Table 2; 0.39 [0.06 to 0.72] for partly skilled/unskilled childhood social class and 1.08 [0.67 to 1.49] for no education age 69). The most socioeconomically disadvantaged by each measure of SEP, adjusted for all other measures of SEP, experienced an additional 0.39 (childhood social class), 0.83 (adult social class), and 1.08 conditions (adult education) at age 69 years (Table 2).

Overall, multimorbidity increased steadily over adulthood with an accelerated accumulation in older ages [from 0.08 (95% CI 0.07 to 0.09) conditions/year between ages 36 and 43 years to 0.19 (0.18 to 0.20) conditions/year between ages 63 and 69 years (linear spline mixedeffects models, Model 1; Table G in S1 File and Fig B in S1 File)]. Women had slightly higher rates of multimorbidity than men between ages 36 and 53 years, after which there were no further differences in rates by sex (Fig C in S1 File). Further, we observed stark socioeconomic inequalities in multimorbidity trajectories, which increased and widened progressively with age. Between ages 43 to 69 years, disadvantaged adulthood social class groups like manual and partly skilled/unskilled individuals experienced greater acceleration of multimorbidity trajectories (Model 3; Table G in S1 File and Fig 4). For instance, compared to professional/intermediate individuals, manual workers experienced an excess of 0.18 conditions and partly skilled/ unskilled individuals 0.31 conditions between ages 36 and 43 years, which increased to 0.30 conditions and 0.49 conditions, respectively between 63 and 69 years (Model 3; Table G in



Fig 4. Socioeconomic inequalities in multimorbidity trajectories by adulthood social class in 3,723 participants from the 1946 MRC NSHD. Note: Fig 4 is based on Model 3 in Table 6 and adjusted for sex, childhood social class, and educational level. NSHD, National Survey of Health and Development.

https://doi.org/10.1371/journal.pmed.1003775.g004

S1 File). The independent (of adult SEP) disparities in multimorbidity trajectories by childhood social class emerged at age 53 years and widened over the remaining period of follow-up (Model 2; Table G in S1 File and Fig.5). In this case, rather than a graded effect by childhood social class, professional and nonmanual childhood social classes both followed a more favourable trajectory than manual and unskilled social classes.

Discussion

Through following the same individuals in the longest running British birth cohort, we report several novel findings. Firstly, we show that multimorbidity accumulation accelerates with age. Secondly, we found that childhood and adulthood SEP, and achieved education, were each independently associated with an increased risk of multimorbidity burden at each age. Thirdly, we demonstrate that adverse SEP is associated with a more rapid acceleration of multimorbid-ity accumulation, providing strong evidence for widening inequalities with age. In sum, socio-economically disadvantaged individuals have both earlier onset and more rapid accumulation of multimorbidity, with independent contributions from both childhood and adulthood SEP.

The few longitudinal studies that have examined associations between SEP and multimorbidity found that low income, low education, low occupational position, and greater area deprivation were associated with increased risk for multimorbidity [10,16,21,22,35,36]. In addition to a shorter follow-up time, previous studies have focused on certain conditions [35] (like cardiometabolic diseases) or analysed disease-specific trajectories [36]. The Whitehall and Twenty-07 studies both examined socioeconomic inequalities in multimorbidity development in the United Kingdom, but the former was restricted to participants aged over 50 in a



Fig 5. Socioeconomic inequalities in multimorbidity trajectories by childhood social class in 3,723 participants from the 1946 MRC NSHD. Note: Fig 5 is based on Model 2 in Table 6 and adjusted for sex, adulthood social class, and educational level. NSHD, National Survey of Health and Development.

https://doi.org/10.1371/journal.pmed.1003775.g005

specific nonpopulation representative sample, while the latter reported trajectories across adulthood in a regional cohort and not in the same individuals [24,25]. A recent study on the Aberdeen birth cohort found that lower social class at birth was associated with higher odds of multimorbidity in middle age, which was partially mediated by educational attainment [22]. However, this study examined associations with multimorbidity at only one time point in middle age (50 to 56 years). Importantly, no previous study has examined socioeconomic patterning of multimorbidity trajectories over the lifecourse taking into account the independent effects of childhood and adulthood socioeconomic circumstances. Previous studies, for example, using electronic health records in Finland, have also reported that inequalities increase with age [10,21]. However, these may be misleading as ascertainment of chronic conditions, most of which are asymptomatic, in some cases for several years, may be biased towards either those with preexisting morbidity or healthier participants who seek healthcare. In capturing data on conditions from multiple sources, including self-report and standardised biomarker assessment on the whole cohort at multiple time points for the same individual, we overcome some concerns on biased ascertainment.

In light of secular trends in health and life expectancy, and in widening childhood and adulthood socioeconomic inequalities, these findings have significant implications for more recent cohorts who face much higher levels of multimorbidity as they age [2]. The consequence of greater prevalence of earlier life obesity and mental ill-health and their sequelae [37], but also enhanced detection and lower thresholds for medication for conditions such as diabetes and hypertension and increased survival from life threatening conditions such as myocardial infarction and some cancers, all of which potentially result in more years lived with multimorbidity.

Strengths and limitations

This study utilises data on individuals followed up from ages 36 to 69 years enabling us to investigate the development of and increasing inequalities in multimorbidity trajectories using robust longitudinal modelling. Additionally, this study uses both periodical cross-sectional and longitudinal analyses to investigate socioeconomic patterning in multimorbidity. The NSHD was a nationally representative sample of all births in 1 week in the UK, with a >95%response rate. As with most longitudinal studies, the NSHD suffers from attrition. Nonetheless, this study included 70% of the original birth cohort (N = 5,362) and is reasonably representative of the British born population of this generation (and with a socioeconomic distribution and mortality rate similar to that observed nationally for this generation) [38,39]. Missing data were addressed using multiple imputation, which is a routinely used robust method. While imputation relies on the MAR assumption, which is not empirically verifiable, we increased the plausibility of the MAR assumption by including socioeconomic variables (which are known predict nonresponse or missingness) in the imputation model. This should help predict missing data with greater precision and minimising nonrandom variation in the values [40]. The NSHD is composed of only members of white European heritage, and findings may not be generalizable to the non-white British population of the same generation. Some conditions like coronary heart disease, stroke, and kidney disorders were defined by only self-report of a doctor's diagnosis. However, this should not affect estimation of multimorbidity, as self-reporting of doctor's diagnoses was not found to bias findings in other studies [41,42], and in NSHD, >90% agreement for diagnosis of diabetes when comparing selfreport with primary care diagnosis was shown [43]. The population-based nature of the study resulted in a comprehensive set of 18 conditions over time; however, studies based on electronic health records usually consist of a larger number of conditions [10,21]. The inclusion of a limited range of common conditions and excluding more rare diseases in mid-adulthood is likely to have led to some underestimation of the extent of multimorbidity. Information on certain conditions (for instance, stroke, diabetes, and arthritis) that mostly occur in older ages was not collected in early adulthood ages, for instance, at age 36 years. However, it is possible that a small number of participants could have experienced early onset of these conditions. The definitions and criteria for diagnoses of various diseases have changed over time, making it difficult to study some conditions in greater detail, for example, all respiratory conditions had to be grouped together due to changing measurement methods and definitions over the last few decades. While there are several definitions of multimorbidity, we chose to use disease counts, which is the most commonly used definition of multimorbidity enabling comparison to other studies, and appropriate as multimorbidity is the outcome of interest (more complex definitions like weighted measures and indices are better suited to studying multimorbidity in relation to outcomes like mortality and healthcare usage) [44]. There are other important dimensions of SEP shown to be related to multimorbidity in older ages such as income and health literacy, which we were unable to take into consideration [24,45]. Lastly, we cannot exclude the possibility of residual confounding by other factors (for instance, family history of disease), which we were unable to control for, which limits causal inferences we can draw about lifetime SEP and multimorbidity trajectories.

Clinical and policy implications

The increase in multimorbidity in older ages is both a natural consequence of longer life expectancies and significant change in lifestyle seen in recent decades. By age 53 years, 52% of our study population experienced multimorbidity, and we observe substantial inequalities in multimorbidity development. The largest increases in conditions over time were observed for those part of the cardiometabolic syndrome (obesity, hypertension, dyslipidaemia, and diabetes), and these conditions were more strongly interconnected over time as demonstrated by the network analysis. Primary care physicians are the main point of contact and ongoing care for patients with multimorbidity, with additional input from geriatricians with expertise in dealing with multimorbidity for older patients, and organ-specific clinical specialists for patients with specific conditions on a single disease basis. While there are limited clinical guidelines for addressing multimorbidity in the UK, these as yet do not include any guidance on targeted care for disadvantaged individuals at higher risk for greater multimorbidity as is clearly shown in this study [46].

The independent and long-lasting shadow of childhood SEP on both the experience and the rate of accumulation of multimorbidities in later life highlights the need for interventions early in life to minimise inequalities in later life health. Importantly, the ongoing and additional inequities observed in relation to adulthood socioeconomic factors stress the need for intervention and prevention efforts starting in childhood, but also their continued provision through the lifecourse to reduce these age-widening inequities in health and their corresponding consequences for healthy ageing, years lived with disability, and reduced life expectancy across all of society.

By 2035, the proportion of the UK population living with 4 or more conditions is expected to increase to 17% from 9.8% in 2015 [47]. Hypertension, chronic pain, obesity, depression, and anxiety are expected to be the leading contributors to multimorbidity, indicating the heterogeneity of diseases likely to co-occur and the strong interlinking between mental and physical health conditions [3]. Given that conditions like obesity, depression, and diabetes are socioeconomically patterned and more prevalent in younger generations, and combined with the projected increase in older age groups due to increasing life expectancies (those >75 years is expected to double to 13% over the next 20 years), the issue of multimorbidity, and its inequalities, will be a heavy burden that poses serious challenges for healthcare systems. Multimorbidity is linked to increased healthcare utilisation, which is largely attributable to the number of interventions required to deal with multiple chronic health conditions in individuals [48]. In addition to placing a substantial economic burden on the healthcare system, it will also require significantly more healthcare professionals and potential restructuring of healthcare provision in the near future to effectively address the issue.

The synergistic clustering of diseases at both the population and individual level is one of the mechanisms by which health inequalities develop and perpetuate in sections of the population [15,49,50]. Hence, any model developed to tackle multimorbidity and to reduce inequalities associated with socioeconomic circumstances should incorporate societal and individual-level factors. Additionally, public health policies that aim to reduce multimorbidity need to start earlier in the lifecourse, must be universal but proportional with increased targeting towards the more socioeconomically disadvantaged population. Individuals with multimorbidity living in deprived areas receive poorer healthcare (for example, shorter consultation times and lower perceived GP empathy) compared to their counterparts in affluent areas [51]. This reflects the inverse care law, which states that the availability of good medical care varies inversely with the need for it in the population served [52]. Hence, there also needs to be a focus on improving healthcare access and services at the ecological level as there is a greater concentration of individuals with multimorbidity among the socioeconomically disadvantaged.

Final summary

In conclusion, this study shows marked and independent effects of childhood and adulthood socioeconomic circumstances on onset, rate of accumulation, and burden of multimorbidity.

These inequalities widen with time and the effects of socioeconomic disadvantage in childhood on multimorbidity emerges in late adulthood and early old age. Strikingly, the most socioeconomically disadvantaged by each measure of SEP experienced an additional 0.39 conditions (childhood social class), 0.83 (adult social class), and 1.08 conditions (adult education) at age 69 years. Multimorbidity is perhaps inevitable in ageing societies with increasing life expectancies, but the inequalities as demonstrated by our study are stark. This calls not only for better access and delivery of healthcare for the more vulnerable, but also for population-based interventions in early life and through the lifecourse that reduce the impact of childhood inequalities if we are truly to address the societal impact of multimorbidity.

Supporting information

S1 STROBE checklist. STROBE Statement. (PDF)

S1 File. Table A. The 18 conditions contributing to the multimorbidity score, with data sources and definitions. Table B. Example of a Delphi form circulated among 10 clinicians for consulting on how to treat the chronicity of each condition. Table C. Prevalence (%) of conditions and multimorbidity at each of the 5 age sweeps in participants from the 1946 MRC NSHD (nonimputed data). Table D. Descriptive characteristics of 3,723 participants from the 1946 MRC NSHD by socioeconomic indicators of interest. Table E. Prevalence of health conditions across the 5 age sweeps by sex in 3,723 participants from the 1946 MRC NSHD. Table F. Mean number of morbidities (health conditions) by covariates of interest in 3,723 participants from the 1946 MRC NSHD. Table G. Multimorbidity trajectories between adulthood (age 36) and old age (age 69) predicted by linear spline mixed-effects modelling in 3,723 participants from the 1946 MRC NSHD. Table H. Results from linear regression modelling assessing associations between SEP and multimorbidity in 3,723 participants from the 1946 MRC NSHD. Table I. Multimorbidity trajectories and rates in change between adulthood (age 36) and old age (age 69) predicted by linear spline mixed-effects modelling in 3,723 participants from the 1946 MRC NSHD—Models without interactions. Fig A. Results from linear regression modelling assessing associations between SEP and multimorbidity in 3,723 participants from the 1946 MRC NSHD. Fig B. Population-based predicted trajectory for multimorbidity from linear spline mixed-effects modelling in 3,723 participants from the 1946 MRC NSHD. Fig C. Population-based predicted trajectories for multimorbidity by sex from linear spline mixed-effects modelling in 3,723 participants from the 1946 MRC NSHD. NSHD, National Survey of Health and Development; SEP, socioeconomic position. (DOCX)

Acknowledgments

We are grateful to the cohort members in the 1946 National Study of Health and Development who have provided data for research throughout their lives for their continuing support. We would like to thank the following clinicians who provided consultation and feedback: James Allinson, Gaby Captur, Sophie Eastwood, Samuel Searle, David Thompson, and Alex Tsui.

We also thank members of the NSHD scientific and data collection teams and NHS Digital for the Hospital Episodes Statistics data.

We did not involve patients or the public in the design, or conduct, or reporting, or dissemination plans of our research.

Results will be disseminated to study participants via the study newsletter and study website.

Author Contributions

Conceptualization: Nishi Chaturvedi, Praveetha Patalay.

Data curation: Amal R. Khanolkar.

Formal analysis: Amal R. Khanolkar, Valerie Kuan.

Methodology: Amal R. Khanolkar, Daniel Davis, Alun Hughes, Marcus Richards, David Bann, Praveetha Patalay.

Project administration: Praveetha Patalay.

Visualization: Amal R. Khanolkar.

Writing - original draft: Amal R. Khanolkar.

Writing – review & editing: Amal R. Khanolkar, Nishi Chaturyedi, Valerie Kuan, Daniel Davis, Alun Hughes, Marcus Richards, David Bann, Praveetha Patalay.

References

- 1. Academy of Medical Sciences. Multimorbidity: a priority for global research. April, 2018. 2018.
- Barnett K, Mercer SW, Norbury M, Watt G, Wyke S, Guthrie B. Epidemiology of multimorbidity and implications for health care, research, and medical education: a cross-sectional study. Lancet. 2012; 380(9836):37–43. https://doi.org/10.1016/S0140-6736(12)60240-2 WOS:000306115900031. PMID: 22579043
- Pearson-Stuttard J, Ezzati M, Gregg EW. Multimorbidity-a defining challenge for health systems. Lancet Public Health. 2019; 4(12):E599–E600. WOS:000500947300005. https://doi.org/10.1016/S2468-2667(19)30222-1 PMID: 31812234
- Walker AE. Multiple chronic diseases and quality of life: patterns emerging from a large national sample, Australia. Chronic Illn. 2007; 3(3):202–18. Epub 2007/12/18. <u>https://doi.org/10.1177/</u> 1742395307081504 PMID: 18083677.
- Peters M, Kelly L, Potter CM, Jenkinson C, Gibbons E, Forder J, et al. Quality of life and burden of morbidity in primary care users with multimorbidity. Patient Relat Outcome Meas. 2018; 9:103–13. https:// doi.org/10.2147/PROM.S148358 WOS:000425357400001. PMID: 29497339
- Jindai K, Nielson CM, Vorderstrasse BA, Quinones AR. Multimorbidity and Functional Limitations Among Adults 65 or Older, NHANES 2005–2012. Prev Chronic Dis. 2016; 13. ARTN 160174 <u>https://doi.org/10.5888/pcd13.160174</u> WOS:000393109300001. PMID: 27809419
- Sum G, Salisbury C, Koh GCH, Atun R, Oldenburg B, McPake B, et al. Implications of multimorbidity patterns on health care utilisation and quality of life in middle-income countries: cross-sectional analysis. J Glob Health. 2019; 9(2). ARTN 020413 https://doi.org/10.7189/jogh.09.020413 WOS:000486599800020. PMID: 31448114
- Cabral GG, de Souza ACD, Barbosa IR, Jerez-Roig J, Souza DLB. Multimorbidity and Its Impact on Workers: A Review of Longitudinal Studies. Saf Health Work. 2019; 10(4):393–9. <u>https://doi.org/10.1016/j.shaw.2019.08.004 WOS</u>:000503797900001. PMID: 31890321
- Agborsangaya CB, Lau D, Lahtinen M, Cooke T, Johnson JA. Health-related quality of life and healthcare utilization in multimorbidity: results of a cross-sectional survey. Qual Life Res. 2013; 22(4):791–9. https://doi.org/10.1007/s11136-012-0214-7 WOS:000318311000011. PMID: 22684529
- Schafer I, Hansen H, Schon G, Hofels S, Altiner A, Dahlhaus A, et al. The influence of age, gender and socio-economic status on multimorbidity patterns in primary care. first results from the multicare cohort study. BMC Health Serv Res. 2012; 12. Artn 89 https://doi.org/10.1186/1472-6963-12-89 WOS:000303809400001. PMID: 22471952
- Singer L, Green M, Rowe F, Ben-Shlomo Y, Morrissey K. Social determinants of multimorbidity and multiple functional limitations among the ageing population of England, 2002–2015. SSM Popul Health 2019; 8. ARTN 100413 https://doi.org/10.1016/j.ssmph.2019.100413 WOS:000498896300039. PMID: 31194123
- Salisbury C, Johnson L, Purdy S, Valderas JM, Montgomery AA. Epidemiology and impact of multimorbidity in primary care: a retrospective cohort study. Br J Gen Pract. 2011; 61(582). https://doi.org/10. 3399/bjgp11X548929 WOS:000288365700006. PMID: 21401985

- Frolich A, Ghith N, Schiotz M, Jacobsen R, Stockmarr A. Multimorbidity, healthcare utilization and socioeconomic status: A register-based study in Denmark. PLoS ONE. 2019; 14(8). ARTN e0214183 https://doi.org/10.1371/journal.pone.0214183 WOS:000484987900005. PMID: 31369580
- Canizares M, Hogg-Johnson S, Gignac MAM, Glazier RH, Badley EM. Increasing Trajectories of Multimorbidity Over Time: Birth Cohort Differences and the Role of Changes in Obesity and Income. J Gerontol B Psychol Sci Soc Sci. 2018; 73(7):1303–14. https://doi.org/10.1093/geronb/gbx004 WOS:000448381600021. PMID: 28199711
- Pathirana TI, Jackson CA. Socioeconomic status and multimorbidity: a systematic review and metaanalysis. Aust N Z J Public Health. 2018; 42(2):186–94. https://doi.org/10.1111/1753-6405.12762 WOS:000429410700013. PMID: 29442409
- Jackson CA, Dobson A, Tooth L, Mishra GD. Body mass index and socioeconomic position are associated with 9-year trajectories of multimorbidity: A population-based study. Prev Med. 2015; 81:92–8. https://doi.org/10.1016/j.ypmed.2015.08.013 WOS:000368421300014. PMID: 26311587
- Roberts KC, Rao DP, Bennett TL, Loukine L, Jayaraman GC. Prevalence and patterns of chronic disease multimorbidity and associated determinants in Canada. Health Promot Chronic Dis Prev Can. 2015; 35(6). https://doi.org/10.24095/hpcdp.35.6.01 WOS:000362426300001. PMID: 26302227
- van den Akker M, Buntinx F, Metsemakers JFM, Roos S, Knottnerus JA. Multimorbidity in general practice: Prevalence, incidence, and determinants of co-occurring chronic and recurrent diseases. J Clin Epidemiol. 1998; 51(5):367–75. https://doi.org/10.1016/s0895-4356(97)00306-5 WOS:000073290300001. PMID: 9619963
- Marmot M. Social determinants of health inequalities. Lancet. 2005; 365(9464):1099–104. <u>https://doi.org/10.1016/S0140-6736(05)71146-6 WOS:000227731800034</u>. PMID: 15781105
- Khanolkar A, Vagero D, Koupil I. Social determinants of cardiac disease biomarkers: investigating a Swedish male cohort at ages 50 and 70. Eur J Prev Cardiol. 2012; 19(3):523–33. Epub 2011/04/01. https://doi.org/10.1177/1741826711405952 PMID: 21450563.
- Kivimaki M, Batty GD, Pentti J, Shipley MJ, Sipila PN, Nyberg ST, et al. Association between socioeconomic status and the development of mental and physical health conditions in adulthood: a multi-cohort study. Lancet Public Health. 2020; 5(3):E140–E9. https://doi.org/10.1016/S2468-2667(19)30248-8 WOS:000518417000009. PMID: 32007134
- Johnston MC, Black C, Mercer SW, Prescott GJ, Crilly MA. Impact of educational attainment on the association between social class at birth and multimorbidity in middle age in the Aberdeen Children of the 1950s cohort study. BMJ Open. 2019; 9(1):e024048. Epub 2019/01/31. https://doi.org/10.1136/ bmjopen-2018-024048 PMID: 30696675; PubMed Central PMCID: PMC6352766.
- Sakib MN, Shooshtari S, St John P, Menec V. The prevalence of multimorbidity and associations with lifestyle factors among middle-aged Canadians: an analysis of Canadian Longitudinal Study on Aging data. BMC Public Health. 2019; 19. ARTN 243 https://doi.org/10.1186/s12889-019-6567-x WOS:000460041100011. PMID: 30819126
- Dugravot A, Fayosse A, Dumurgier J, Bouillon K, Rayana TB, Schnitzler A, et al. Social inequalities in multimorbidity, frailty, disability, and transitions to mortality: a 24-year follow-up of the Whitehall II cohort study. Lancet Public Health. 2020; 5(1):e42–e50. Epub 2019/12/16. https://doi.org/10.1016/S2468-2667(19)30226-9 PMID: 31837974; PubMed Central PMCID: PMC7098476.
- Katikireddi SV, Skivington K, Leyland AH, Hunt K, Mercer SW. The contribution of risk factors to socioeconomic inequalities in multimorbidity across the lifecourse: a longitudinal analysis of the Twenty-07 cohort. BMC Med. 2017; 15(1):152. Epub 2017/08/25. https://doi.org/10.1186/s12916-017-0913-6 PMID: 28835246; PubMed Central PMCID: PMC5569487.
- Mondor L, Cohen D, Khan AI, Wodchis WP. Income inequalities in multimorbidity prevalence in Ontario, Canada: a decomposition analysis of linked survey and health administrative data. Int J Equity Health. 2018; 17(1):90. Epub 2018/06/27. https://doi.org/10.1186/s12939-018-0800-6 PMID: 29941034; PubMed Central PMCID: PMC6019796.
- Galobardes B, Smith GD, Lynch JW. Systematic review of the influence of childhood socioeconomic circumstances on risk for cardiovascular disease in adulthood. Ann Epidemiol. 2006; 16(2):91–104. Epub 2005/11/01. https://doi.org/10.1016/j.annepidem.2005.06.053 PMID: 16257232.
- Harper S, Lynch J, Smith GD. Social determinants and the decline of cardiovascular diseases: understanding the links. Annu Rev Public Health. 2011; 32:39–69. Epub 2011/01/12. https://doi.org/10.1146/ annurev-publhealth-031210-101234 PMID: 21219168.
- 29. Wadsworth M, Kuh D, Richards M, Hardy R. Cohort profile: The 1946 National Birth Cohort (MRC National Survey of Health and Development). Int J Epidemiol. 2006; 35(1):49–54. <u>https://doi.org/10.1093/ije/dyi201</u> WOS:000235281700013. PMID: <u>16204333</u>

- Melzer DCD, J. Winder R, Masoli J, Richards S, Ble A. The Age UK almanac of disease profiles in later life—A reference on the frequency of major diseases, conditions and syndromes affecting older people in England. 2015.
- Dalkey N, Helmer O. An Experimental Application of the Delphi Method to the Use of Experts. Manag Sci. 1963; 9(3):458–67. https://doi.org/10.1287/mnsc.9.3.458 WOS:A1963CBG1200008.
- Howe LD, Tilling K, Matijasevich A, Petherick ES, Santos AC, Fairley L, et al. Linear spline multilevel models for summarising childhood growth trajectories: A guide to their application using examples from five birth cohorts. Stat Methods Med Res. 2016; 25(5):1854–74. https://doi.org/10.1177/ 0962280213503925 WOS:000385555400008. PMID: 24108269
- Collins LM, Schafer JL, Kam CM. A comparison of inclusive and restrictive strategies in modern missing data procedures. Psychol Methods. 2001; 6(4):330–51. Epub 2002/01/10. PMID: 11778676.
- 34. Mostafa T, Narayanan M, Pongiglione B, Dodgeon B, Goodman A, Silverwood RJ, et al. Improving the plausibility of the missing at random assumption in the 1958 British birth cohort: A pragmatic data driven approach: UCL Centre for Longitudinal Studies; 2020. Available from: https://cls.ucl.ac.uk/wp-content/uploads/2020/04/CLS-working-paper-2020-6-Improving-the-plausibility-of-the-missing-at-random-assumption-in-the-1958-British-birth-cohort.pdf.
- 35. Singh-Manoux A, Fayosse A, Sabia S, Tabak A, Shipley M, Dugravot A, et al. Clinical, socioeconomic, and behavioural factors at age 50 years and risk of cardiometabolic multimorbidity and mortality: A cohort study. PLoS Med. 2018; 15(5). ARTN e1002571 https://doi.org/10.1371/journal.pmed.1002571 WOS:000434236000010. PMID: 29782486
- Hsu HC. Trajectories of multimorbidity and impacts on successful aging. Exp Gerontol. 2015; 66:32–8. Epub 2015/04/15. https://doi.org/10.1016/j.exger.2015.04.005 PMID: 25871727.
- Patalay P, Gage SH. Changes in millennial adolescent mental health and health-related behaviours over 10 years: a population cohort comparison study. Int J Epidemiol. 2019; 48(5):1650–64. Epub 2019/ 03/01. https://doi.org/10.1093/ije/dyz006 PMID: 30815691; PubMed Central PMCID: PMC6904321.
- Pierce MB, Silverwood RJ, Nitsch D, Adams JE, Stephen AM, Nip W, et al. Clinical Disorders in a Post War British Cohort Reaching Retirement: Evidence from the First National Birth Cohort Study. PLoS ONE. 2012; 7(9). ARTN e44857 https://doi.org/10.1371/journal.pone.0044857 WOS:000309388400044. PMID: 23028647
- Stafford M, Black S, Shah I, Hardy R, Pierce M, Richards M, et al. Using a birth cohort to study ageing: representativeness and response rates in the National Survey of Health and Development. Eur J Ageing. 2013; 10(2):145–57. Epub 2013/05/03. https://doi.org/10.1007/s10433-013-0258-8 PMID: 23637643; PubMed Central PMCID: PMC3637651.
- 40. Sterne JA, White IR, Carlin JB, Spratt M, Royston P, Kenward MG, et al. Multiple imputation for missing data in epidemiological and clinical research: potential and pitfalls. BMJ. 2009; 338:b2393. Epub 2009/ 07/01. https://doi.org/10.1136/bmj.b2393 PMID: 19564179; PubMed Central PMCID: PMC2714692.
- Tisnado DM, Adams JL, Liu HH, Damberg CL, Chen WP, Hu FA, et al. What is the concordance between the medical record and patient self-report as data sources for ambulatory care? Med Care. 2006; 44(2):132–40. https://doi.org/10.1097/01.mlr.0000196952.15921.bf WOS:000234950400006. PMID: 16434912
- Klungel OH, de Boer A, Paes AHP, Seidell JC, Bakker A. Cardiovascular diseases and risk factors in a population-based study in The Netherlands: agreement between questionnaire information and medical records. Neth J Med. 1999; 55(4):177–83. https://doi.org/10.1016/s0300-2977(99)00045-5
 WOS:000083135700003. PMID: 10555434
- Pastorino S, Richards M, Hardy R, Abington J, Wills A, Kuh D, et al. Validation of self-reported diagnosis of diabetes in the 1946 British birth cohort. Prim Care Diabetes. 2015; 9(5):397–400. Epub 2014/07/20. https://doi.org/10.1016/j.pcd.2014.05.003 PMID: 25037637; PubMed Central PMCID: PMC4582042.
- Huntley AL, Johnson R, Purdy S, Valderas JM, Salisbury C. Measures of Multimorbidity and Morbidity Burden for Use in Primary Care and Community Settings: A Systematic Review and Guide. Ann Fam Med. 2012; 10(2):134–41. <u>https://doi.org/10.1370/afm.1363</u> WOS:000301996400007. PMID: 22412005
- Stirland LE, Gonzalez-Saavedra L, Mullin DS, Ritchie CW, Muniz-Terrera G, Russ TC. Measuring multimorbidity beyond counting diseases: systematic review of community and population studies and guide to index choice. BMJ. 2020; 368:m160. Epub 2020/02/20. <u>https://doi.org/10.1136/bmj.m160</u> PMID: 32071114.
- 46. Farmer C, Fenu E, O'Flynn N, Guthrie B. GUIDELINES Clinical assessment and management of multimorbidity: summary of NICE guidance. BMJ. 2016; 354. ARTN i4843 <u>https://doi.org/10.1136/bmj.i4843</u> WOS:000384319400002. PMID: 27655884
- 47. Kingston A, Robinson L, Booth H, Knapp M, Jagger C, Project M. Projections of multi-morbidity in the older population in England to 2035: estimates from the Population Ageing and Care Simulation

(PACSim) model. Age Ageing. 2018; 47(3):374–80. https://doi.org/10.1093/ageing/afx201 WOS:000432043700011. PMID: 29370339

- Wolff JL, Starfield B, Anderson G. Prevalence, expenditures, and complications of multiple chronic conditions in the elderly. Arch Intern Med. 2002; 162(20):2269–76. https://doi.org/10.1001/archinte.162.20. 2269 WOS:000179148300001. PMID: 12418941
- Tsai AC, Venkataramani AS. Syndemics and Health Disparities: A Methodological Note. AIDS Behav. 2016; 20(2):423–30. Epub 2015/12/15. https://doi.org/10.1007/s10461-015-1260-2 PMID: 26662266; PubMed Central PMCID: PMC4755906.
- Mendenhall E, Kohrt BA, Norris SA, Ndetei D, Prabhakaran D. Non-communicable disease syndemics: poverty, depression, and diabetes among low-income populations. Lancet. 2017; 389(10072):951–63. WOS:000395263400035. https://doi.org/10.1016/S0140-6736(17)30402-6 PMID: 28271846
- Mercer SW, Zhou Y, Humphris GM, McConnachie A, Bakhshi A, Bikker A, et al. Multimorbidity and Socioeconomic Deprivation in Primary Care Consultations. Ann Fam Med. 2018; 16(2):127–31. Epub 2018/03/14. https://doi.org/10.1370/afm.2202 PMID: 29531103; PubMed Central PMCID: PMC5847350.
- Mercer SW, Watt GC. The inverse care law: clinical primary care encounters in deprived and affluent areas of Scotland. Ann Fam Med. 2007; 5(6):503–10. Epub 2007/11/21. https://doi.org/10.1370/afm. 778 PMID: 18025487; PubMed Central PMCID: PMC2094031.